Author’s response to reviews

**Title:** Determinants of Severe Acute Malnutrition Among Under Five Children In rural Enebsie Sarmidr District, East Gojjam Zone, North West Ethiopia, 2016

**Authors:**

Abate Awoke (abateawoke@yahoo.com)

Mulatu Ayana (mulatuayana@gmail.com)

Tenaw Gualu (tenawgualu@yahoo.com)

**Version:** 2 **Date:** 12 May 2017

**Author’s response to reviews:**

I found the comment worth enough. The following corrections have been taken according to the reviewers comment

1. Some paragraphs of the introduction have been re written with updated references in a way that they are focused to the topic of interest

2. In the methodology(detail description of the study area, sampling procedure, study design, tool preparation...etc have been included)

3. The tables of the result have been rewritten with description of the main findings of the table. One paragraph is removed which is not worth enough

4. The references list have been updated and rewritten as new references have been used.

Generally many things have been modified according to the comment.